Author's response to reviews

Title: The estimated economic burden of genital herpes in the United States. An analysis using two costing approaches

Authors:

Thomas D Szucs (Thomas.Szucs@hirslanden.ch)
Karin Berger (merg.munich@t-online.de)
David N Fisman (dfisman@caregroup.harvard.edu)
Stephan Harbarth (harbarth@post.harvard.edu)

Version: 7 Date: 28 Jun 2001

Dear Katy:

Please find attached the final version of the MS, including your editorial advice and some additional requests on email addresses and competing interests.

Please, let me know if you have any trouble again reading the references.

Thanks again for considering our MS.

Yours, Stephan Harbarth